Author's response to reviews

Title: Homeopathy for mental fatigue: lessons from a randomized, triple blind, placebo-controlled cross-over clinical trial

Authors:

Michael E Dean (medean@tiscali.co.uk)
Raj Karsandas (rajkarsandas@hotmail.com)
J Martin Bland (martin.bland@york.ac.uk)
Debbie Gooch (d.gooch@psych.york.ac.uk)
Hugh MacPherson (hugh.macpherson@york.ac.uk)

Version: 2 Date: 30 March 2012

Author's response to reviews:

Amended to include tables in main manuscript.